Author's response to reviews

Title: Chromosome 7p linkage and association study for diabetes related traits and type 2 diabetes in an African-American population enriched for nephropathy

Authors:

Tennille S Leak (leakt@edc.pitt.edu)
Carl D Langefeld (clangefe@wfubmc.edu)
Keith L Keene (klk4a@virginia.edu)
Carla J Gallagher (cgallagher@hmc.psu.edu)
Lingyi Lu (llu@wfubmc.edu)
Josyf C Mychaleckyj (jmychale@virginia.edu)
Stephen S Rich (ssr4n@virginia.edu)
Barry I Freedman (bfreedma@wfubmc.edu)
Donald W Bowden (dbowden@wfubmc.edu)
Michele M Sale (msale@virginia.edu)

Version: 5 Date: 28 January 2010

Author's response to reviews:

Corrected proof